Author’s response to reviews

Title: The impact of cancer on diabetes outcomes.

Authors:
Anne Arreskov (anne.arreskov@sund.ku.dk)
Maria Olsen (maria@aahlander.dk)
Sandra Pouplier (sandra.pouplier@gmail.com)
Volkert Siersma (siersma@sund.ku.dk)
Christen Andersen (christenla@gmail.com)
Søren Friis (friis@CANCER.DK)
Niels Olivarius (olivarius@sund.ku.dk)

Version: 3 Date: 23 Apr 2019

Author’s response to reviews:

Dear Editor

Thank you very much for having read the article again.

According to the text overlap, it concerns mainly the “method” paragraph, which is important to understand the study. As the RCT has given occasion for several publications, the method has also been described in several other publications, which we also have reference to. We have now changed some of the text a bit and further inserted references to:


The Ethics committee approved Verbal Consent to Participate – this is also stated in the “Ethics approval”

We have attached a “clean version” of the manuscript: “Main Document_R3_clean”.

We look forward to hearing from you again.

On behalf of all authors.

Yours sincerely

Anne Arreskov

M.D. Ph.D student
Research Unit for General Practice in Copenhagen
University of Copenhagen